Author’s response to reviews

Title: Epidemiology of Acute Kidney Injury in Hospitalized Pregnant Women in China

Authors:

Diankun Liu (tennisldk@gmail.com)
Wenjuan He (1021423855@qq.com)
Yanqin Li (383882351@qq.com)
Mengqi Xiong (952726393@qq.com)
Long Wang (13275860963@163.com)
Jingxin Huang (wongchingyan3@163.com)
Ling Jia (dlps99@163.com)
Shuling Yuan (yuan19456@163.com)
Sheng Nie (niesheng0202@126.com)

Version: 2 Date: 09 Feb 2019

Author’s response to reviews:

Please provide the figure titles/legends under a separate heading of 'Figure Legends' after the References.

Response: We have provided the figure legends under a separate heading of 'Figure Legends' after the References accordingly.

-- Please also add a section “Supplementary files” where you list the following information about your supplementary material:

- File name
- Title of data
- Description of data

Response: We have provided the information under a separate heading of 'Supplementary files' after the References accordingly.
Please ensure that all additional files are explicitly referred to in the main text.

Response: We have checked all the additional files accordingly.

-- Thank you for providing a response to the reviewers/Cover Letter. As these documents are no longer required at this stage of the publication process, please remove them from your submission’s supplementary files.

Response: We have removed the rebuttal letter from submission's supplementary files.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We have uploaded our manuscript as a clean version accordingly.